Author's response to reviews

Title: Efficacy and safety of moxifloxacin in acute exacerbations of chronic bronchitis: a prospective, multicenter, observational study (AVANTI)

Authors:

Alexander Chuchalin (chuchalin@inbox.ru)
Maryna Zakharova (ZakharovaMarina@i.ua)
Dejan Dokic (drdejand@yahoo.com)
Mahir Tokić (asim_tokic@yahoo.com)
Hans-Peter Marschall (hans-peter.marschall@bayer.com)
Thomas Petri (thomas.petri@bayer.com)

Version: 2 Date: 18 June 2012

Author's response to reviews:

Dear Sir/Madam,

we discovered a small error in the abstract of the submitted manuscript. Instead of 2.3% it should read 2.13% (=57/2672). The abstract in the manuscript as well as on the "manuscript details" page was revised accordingly!

Best regards Thomas Petri